Author's response to reviews

Title: Stress Induced Hypertensive Response: Should it be evaluated more carefully?

Authors:

Nagehan Nagehan Kucukler (drnagehan@gmail.com)
Fatih Yalcin (fyalcin1@jhmi.edu)
Theodore P Abraham (tabraha3@jhmi.edu)
Mario J Garcia (mariogar@montefiore.org)

Version: 3 Date: 11 August 2011

Author's response to reviews:

Aug 11, 2011
Dear Editor,

We are submitting the revised manuscript, entitled “Stress Induced Hypertensive Response: Should it be evaluated more carefully?” for reevaluation in the “Cardiovascular Ultrasound”.

This review has been expanded and divided with subheadings and the figures were added. We have more discussed the usefulness of echocardiography, tissue Doppler imaging, evaluation of myocardial functions, contractility, myocardial perfusion and application of new echo technologies during exercise stress echocardiography in the setting of exercise hypertension. We have additionally better discussed prognostic importance of stress induced hypertension, development of hypertension in the future and plausible treatments. The recommended citations have been addressed and all changes have also been highlighted in the text.

Please consider the review for publication in the “Cardiovascular Ultrasound”.

Sincerely,
Nagehan Kucukler, MD
Johns Hopkins Medical Institutions
Department of Cardiology
720 Rutland Avenue / Ross Room 1044
Baltimore, MD 21205 USA
Phone: (410) 502-2505 Fax: (410) 502-3559
Email: drnagehan@gmail.com